Author's response to reviews

Title: Does treatment of subsyndromal depression improve depression and diabetes related outcomes: protocol for a randomised controlled comparison of psycho-education, physical exercise and treatment as usual

Authors:

Mirjana Pibernik-Okanović (pibernik@idb.hr)
Dea Ajduković (dea.ajdukovic@idb.hr)
Marijana Vučić (vucic@idb.hr)
Norbert Hermanns (hermanns@diabetes-zentrum.de)

Version: 3 Date: 18 January 2011

Author's response to reviews:

Dear Sirs,

Due to a technical error in the automatic numbering of the reference list, there were several errors in the text and the reference list, which have now been corrected. In the present revised manuscript there are 39 references in total, each of them is correctly cited in the text and the list is numbered properly.

With kind regards,
Sincerely,
Mirjana Pibernik-Okanovic, PhD